Author's response to reviews

Title: Do Trialists Endorse Clinical Trial Registration? Survey of a PUBMED Sample.

Authors:

Ludovic Reveiz (lureveiz@colsanitas.com)
Karmela Krleza-Jeric (KKrleza-Jeric@cihr-irsc.gc.ca)
An Wen Chan (anwen.chan@utoronto.ca)
Sylvia de Aguiar (sylvia_deaguiar@hotmail.com)

Version: 2 Date: 2 April 2007

Author’s response to reviews:

We are submitting a new version to comply with the journal style. We included the following sections:
* Include a title page
* Include an Abstract section
* Competing interests declaration
* Authors contributions list
* Acknowledgements list

Many thanks for your help
Best wishes

Ludovic Reveiz